Author's response to reviews

Title: Prevalence of metabolic syndrome in a cohort of Chinese schoolchildren: comparison of two definitions and assessment of adipokines as components by factor analysis

Authors:

- Qiaoxuan Wang (xuanzy.18@163.com)
- Jinhua Yin (yinjihh1@yahoo.com.cn)
- Hong Cheng (chenghong692000@163.com)
- Xiaoyuan Zhao (xyzhao2009@163.com)
- Hongding Xiang (xianghd@vip.sohu.net)
- Lam Hugh Simon (hshslam@cuhk.edu.hk)
- Jie Mi (jiemi@vip.163.com)
- Ming Li (liming@pumch.cn)

Version: 6 Date: 14 February 2013

Author's response to reviews:

Dear editor:

Thank you for giving us the opportunity to revise again. With the assistance of Prof. Simon, who is a native-English speaker with scientific expertise from Prince of Wales Hospital, Hong Kong, we have carefully corrected our language and made some revisions according to the reviewers’ concerns. We hope this improved manuscript will meet the high quality requirements of your journal.

Best wishes

Ming Li
PUMC Hospital